528 5 4
2007 % 11 A

W o% MO
IRON STEEL VANADIUM TITANIUM

Vol.28,No.4
November 2007

~ o,
R LEHEA

Bumer) , 2 #9 £ B8 & T AE SE K LA Bk A0 ik T+
&, EEENMEFREREE RN TZIE
HEE T+ EENER.

REET IR TEIA RS, Wik A5 TR B2

RH - MFB 15 /& 3 12 30 € R $0 &
B 5 ik B 7 R L 3R
SREK, R B, LIE Kk W

(LB I R¥FRE ST, WL E L 063009)

W ERIEXE] RH- MFB AP EE, R CRXRHBLINEHNBRRDERKES TR EREER MH
BERAFEDRBREEN I FERATHR, ARG RIERIEREER, 00 T RS W R E N2
B, EETIZREFHEMNBEROBESBYEE VEEBRRN TELRARERBKEMER,

X7 :RH - MFB; 8B M W3l Bk E

hESES:TF69.4  XMKIRIRIW:A  XEHS 1004 - 7638(2007)04 - 0029 - 07

Numerical Simulation on Molten Steel Flow and Carbon Concentration
Distribution in Ladle during RH Vacuum Degassing Process

Gao Aimin, Liang Juan, Ai Liqun, Zhang Xiang
(School of Metallurgy & Energy, Hebei Polytechnic University, Tangshan 063009, Hebei, China)

Abstract: According to the actual production condition of RH — MFB in one plant, the coupling mathematic
model of molten steel flow and carbon concentration distribution in ladle is established by using the commercial
code CFX. The flow characteristic of molten steel and variation of carbon concentration are investigated, and
the validity of mathematic model is verified by comparing simulation results with measuring results. Carbon
concentration distribution is analyzed at the last stage of decarburization, and the influence of process equip-
ment on decarburization process is discussed. So some useful theoretical evidence instructions are put forward
for ultra — low — carbon steel production.
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Fig.5 Carbon concentration distribution at last stage of decarburization
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Fig.6 Comparison on molten steel velocity at different argon blowing flow
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Fig.7 Variation of average carbon concentration
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